Author's response to reviews

Title: BACH1 Ser919Pro variant and breast cancer risk

Authors:

Pia Vahteristo (pia.vahteristo@helsinki.fi)
Kristiina Yliannala (kristiina.yliannala@ktl.fi)
Anitta Tamminen (anitta.tamminen@hus.fi)
Hannaleena Eerola (hannaleena.eerola@fimnet.fi)
Carl Blomqvist (carl.blomqvist@hus.fi)
Heli Nevanlinna (heli.nevanlinna@hus.fi)

Version: 4 Date: 27 December 2005

Author's response to reviews:

Dear BMC Cancer Editorial Team,

Please find attached the new version of the manuscript "BACH1 Ser919Pro variant and breast cancer risk". All the suggested formatting changes have been made, and the manuscript has been carefully checked for typing errors and such. Please let me know if there are any additional requests. For some reason, references 1, 8, 10, 11 and 23 are still incomplete in the submitted version even though they are correct in the manuscript; this occurred also previously and it was replied to me that it can be fixed by the Editorial Team. Therefore, I'm sending the word-document of the final version of the manuscript also through e-mail.

Yours Sincerely,

Pia Vahteristo